cambridge.org/sus

Research Article

Cite this article: Braun J, Stenzel F, Bodirsky BL, Jalava M, Gerten D (2022). Dietary changes could compensate for potential yield reductions upon global river flow protection. *Global Sustainability* **5**, e14, 1–13. https:// doi.org/10.1017/sus.2022.12

Received: 6 November 2020 Revised: 18 October 2021 Accepted: 29 June 2022

Key words:

agriculture; earth systems (land; water and atmospheric); land use; modeling and simulation

Author for correspondence:

Johanna Braun, E-mail: johanna.braun@pik-potsdam.de

© The Author(s), 2022. Published by Cambridge University Press. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.



Dietary changes could compensate for potential yield reductions upon global river flow protection

Johanna Braun¹ , Fabian Stenzel^{1,2,3} , Benjamin Leon Bodirsky¹ , Mika Jalava⁴ and Dieter Gerten^{1,2,3}

¹Potsdam Institute for Climate Impact Research (PIK), Member of the Leibniz Association, P. O. Box 60 12 03, D-14412 Potsdam, Germany; ²Department of Geography, Humboldt-Universität zu Berlin, Unter den Linden 6, D-10099 Berlin, Germany; ³Integrative Research Institute on Transformations of Human–Environment Systems, Unter den Linden 6, D-10099 Berlin, Germany and ⁴Water and Development Research Group, Aalto University, P. O. Box 15200, FI-00076 Aalto, Finland

Non-technical summary. Globally, freshwater systems are degrading due to excessive water withdrawals. We estimate that if rivers' environmental flow requirements were protected, the associated decrease in irrigation water availability would reduce global yields by \sim 5%. As one option to increase food supply within limited water resources, we show that dietary changes toward less livestock products could compensate for this effect. If all currently grown edible feed was directly consumed by humans, we estimate that global food supply would even increase by 19%. We thus provide evidence that dietary changes are an important strategy to harmonize river flow protection with sustained food supply.

Technical summary. To protect global freshwater ecosystems and restore their integrity, freshwater withdrawals could be restricted to maintain rivers' environmental flow requirements (EFRs). However, without further measures, reduced irrigation water availability would decrease crop yields and put additional pressure on global food provision. By comparing the quantitative effects of both global EFR protection and dietary changes on regional and global food supply in a spatially explicit modeling framework, we show that dietary changes toward less livestock products could effectively contribute to solving this trade-off. Results indicate that protection of EFRs would almost halve current global irrigation water withdrawals and reduce global crop yields by 5%. Limiting animal protein share to 25, 12.5 and 0% of total protein supply and shifting released crop feed to direct human consumption could however increase global food supply by 4, 11 and 19%, respectively. The effects are geographically decoupled: water-scarce regions such as the Middle East, or South and Central Asia would be most affected by EFR protection, whereas dietary changes are most effective in North America and Europe. This underpins the disproportionally high responsibilities of countries with resource-intensive diets and the need for regionally adapted and diverse strategies to transform the global food system toward sustainability.

Social media summary. Combining dietary changes and global river flow protection could contribute to a more sustainable food system.

1. Introduction

The global food system is a dominant driver of environmental change, thereby substantially contributing to transgressions of planetary boundaries (Campbell et al., 2017; Gerten et al., 2020; Gerten & Kummu, 2021; Rockström, Edenhofer, Gaertner, & Declerck, 2020; Willett et al., 2019). Expansion of agricultural land through conversion of natural ecosystems and agricultural intensification through fertilizers, pesticides and irrigation have cumulatively led to biodiversity loss, global warming, soil degradation, eutrophication and pollution at the global scale (Foley et al., 2011). Freshwater ecosystems, such as rivers, lakes and wetlands, are among the most severely threatened habitats worldwide and particularly impacted by agricultural activities. Apart from water pollution and eutrophication through fertilizers and pesticides, an estimated 2158–3185 km³ of water are withdrawn for irrigation every year, corresponding to around 70% of total human water withdrawals (Droppers, Franssen, van Vliet, Nijssen, & Ludwig, 2020 and references therein; FAO, 2020). While ~30-40% of irrigation water requirements are abstracted from groundwater, ~80% of net irrigation abstractions (after accounting for return flows) originate from surface waters (Döll et al., 2012, 2014). With a further growing population and an increasingly bio-based economy, irrigation water demands are likely to increase and further intensify the pressure on riverine ecosystems (Stenzel, Gerten, Werner, & Jagermeyr, 2019; Wada & Bierkens, 2014).

To prevent future detrimental effects on freshwater ecosystems and restore already degraded ecosystems, it has been advocated to restrict human water withdrawals (Arthington et al., 2018; Hogeboom, Bruin, Schyns, Krol, & Hoekstra, 2020; Richter, Davis, Apse, & Konrad, 2012). By setting a water withdrawal cap at the river basin scale, human water appropriation could be kept within ecological boundaries. Aiming at defining ecological water needs, the concept of environmental flow requirements (EFRs) was suggested, that is, the 'quantity, timing, and quality of freshwater flows and levels necessary to sustain aquatic ecosystems which, in turn, support human cultures, economies, sustainable livelihoods, and well-being' (Arthington et al., 2018). While rainfed agriculture has little effects on river flow and can even lead to increases in runoff (Rost et al., 2008), current transgressions of EFRs can largely be attributed to water withdrawals for irrigation (Jägermeyr, Pastor, Biemans, & Gerten, 2017). In an attempt to define dangerous levels of human interference with the water cycle, the planetary boundary for freshwater use is currently defined, among other elements, by these EFRs (Gleeson et al., 2020; Steffen et al., 2015). While international implementations of EFR protection policies could thus help to get back into a safe operating space regarding water withdrawals, they would - all else being equal - unavoidably reduce the amount of available irrigation water and thereby the yields on irrigated cropland. Thus, restricting water use could not only result in economic disadvantages for farmers at the local level (Pang, Li, Sun, Yang, & Yang, 2018), but irrigation water constraints at the global level might even threaten food security (Liu et al., 2017). In order to respect social boundaries (Raworth, 2012), among them sufficient food for everyone, global environmental flow protection measures would thus have to be accompanied by a sustainable increase of food supply within limited water resources. Only then, Sustainable Development Goals addressing both freshwater ecosystems and hunger can be jointly achieved (Jägermeyr et al., 2017; UN, 2015).

Different measures have been proposed to reduce the food system's water footprint or to sustainably increase food supply without increasing water use, among them better farm water management and a reduction of food losses and waste (e.g. Davis, Rulli, Seveso, & D'Odorico, 2017; Jägermeyr et al., 2015; Kummu et al., 2012). More recently, dietary changes toward less animal-based products are receiving increasing attention (Berners-Lee, Kennelly, Watson, & Hewitt, 2018; Jalava, Kummu, Porkka, Siebert, & Varis, 2014; Poore & Nemecek, 2018). Since animal-based products are generally associated with higher water footprints than plant-based products (Hoekstra, 2012), dietary changes toward higher shares of plantbased products might significantly reduce the water consumption of global agriculture or, respectively, increase food supply without additional water use. These potentials can be explained by the unfavorable resource conversion efficiency from plant matter to animal products: to produce 1 calorie of animal product, several calories of feed are needed (Berners-Lee et al., 2018). By reallocating human-edible feed to direct human consumption or by freeing up land for growth of plant-based alternatives to animal products (Foley et al., 2011; Godfray et al., 2010), dietary changes toward less animal-based products are thus recognized as one important cornerstone for achieving future food security vis-à-vis limited resources, among them water.

By now, the potential negative effect of global EFR protection on yields (Bonsch et al., 2015; Jägermeyr et al., 2017; Rosa et al., 2018, 2019) and the potentials to increase calorie supply through dietary changes (Berners-Lee et al., 2018; Cassidy, West, Gerber, & Foley, 2013; Foley et al., 2011; West et al., 2014) have been mostly investigated separately. By combining agro-hydrologic simulations from the dynamic global vegetation and water balance model LPJmL (Schaphoff et al., 2018b) with a calorie accounting scheme based on FAO Food Balance Sheet data (FAOSTAT, 2017), this scenario study quantifies whether and to what extent regional and global dietary changes (assuming different levels of reduction in livestock production) could compensate for yield declines on irrigated cropland that would occur under rigid EFR protection. In other words, instead of analyzing direct effects of dietary changes on agricultural water use, we estimate possible increases in food supply through a reallocation of crop feed to direct human consumption, both from rainfed and irrigated cropland. This approach is based on (here updated and expanded) calculations in Gerten et al. (2020), who estimated potentials from different measures, among them dietary changes, to sustainably increase food supply within planetary boundaries including the freshwater boundary defined by EFR constraints.

2. Methods

To compare the effects of EFR protection and dietary changes on global calorie supply, this study integrates (1) simulations with a global model capable of representing a hypothetical EFR protection scenario and the resulting effects on crop yields with (2) a calorie calculation scheme which converts the crop yields to plant and livestock calorie supply depending on livestock production levels. In the following, we first introduce the LPJmL model used and the two irrigation scenarios considered (current irrigation and reduced irrigation due to EFR protection). Then we describe the calculation scheme used to convert yields to calorie supply based on current dietary habits and three scenarios with incrementally decreased livestock production including a scenario without livestock production at all. Ultimately, this allows us to compare both separate and combined effects of irrigation and diet scenarios on potential global and regional calorie supply.

2.1 The dynamic global vegetation and water balance model LPJmL

LPJmL (here, version 3.5) dynamically simulates the growth and productivity of both natural and (irrigated or rainfed) agricultural vegetation in an internally consistent framework by interconnecting underlying water, carbon and energy fluxes (for a detailed model description and evaluation see Schaphoff et al., 2018a, 2018b). Simulations are performed at daily time steps with a spatial resolution of 0.5°. To represent agricultural production, managed grasslands and 12 crop functional types (CFTs) are specified. Food crops which are not covered by the 12 CFTs are pooled and parametrized as managed grasslands as well (CFT13, 'others'). Assimilated carbon is distributed to four plant organs (roots, leaves, the harvestable storage organ and a pool representing stems and mobile reserves), depending on phenological stage and water availability. While agricultural land-use patterns and the extent of irrigated areas are defined by the land-use input, the required irrigation volume on irrigated cropland is calculated internally based on grid cell- and CFT-specific water requirements and the prevailing irrigation system. For this, daily net irrigation requirements are calculated for each CFT and cell, based on the soil water deficit of the top 50 cm soil layer. Additionally, system-specific inefficiencies are determined for drip, sprinkler and surface irrigation, which take into account the system-specific inefficiencies (geographic distribution and parametrization according to Jägermeyr et al., 2015). The

resulting gross irrigation requirements are requested for withdrawal if soil moisture falls below a CFT-specific threshold, thus determining the daily scheduling of irrigation events. Water withdrawals for household, industry and livestock (drinking and cleansing water) are prescribed according to Flörke et al. (2013) and prioritized over irrigation. In the baseline scenario (no EFR protection), total water withdrawals per grid cell are constrained only by total local availability of renewable freshwater resources (river discharge including groundwater baseflow, lakes and reservoirs). Non-renewable abstractions from groundwater are not considered, but are estimated to constitute <20% of global irrigation water withdrawals (Wada, van Beek, & Bierkens, 2012). River discharge is computed based on accumulated surface and subsurface runoff along the river network (Rost et al., 2008; Schaphoff et al., 2018b); its seasonality is reproduced reasonably well in many regions (Schaphoff et al., 2018a). To account for the impact of reservoirs and dams on discharge as well as irrigation water availability, LPJmL includes a reservoir module (Biemans et al., 2011) with a generic reservoir operation model that differentiates reservoir functions (irrigation vs. other purposes), covering ~7000 dams and reservoirs of the GRanD database (Lehner et al., 2011). The minimum water release from reservoirs to the river is set to 10% of the mean monthly inflow, while the remainder can be diverted to irrigated land if needed to meet a cell's irrigation demand in addition to local discharge.

2.2 Simulation protocol and irrigation scenarios

For this study, grid-cell-specific model outputs from two irrigation scenarios were comparatively analyzed regarding crop yields and irrigation: (1) in a baseline scenario water withdrawals were constrained by total local freshwater availability and (2) in an EFR scenario water withdrawals were additionally constrained by local EFRs (to represent their protection). For both scenarios, land use with geographically explicit distribution of rainfed and irrigated CFTs was held constant at year 2005 level over the simulation period. The respective land-use input is based on the MIRCA 2000 dataset (Portmann, Siebert, & Döll, 2010) with an adapted extent of irrigated areas for 2005 (Siebert et al., 2015) and geographically explicit distribution of irrigation systems (Jägermeyr et al., 2015). To represent regional differences in landuse intensities, averaged national FAO yield statistics for 2000-2009 (FAOSTAT, 2020) were used to calibrate simulated yields for management (for a description of the procedure and land-use dataset, see Fader, Rost, Müller, Bondeau, and Gerten, 2010 and Jägermeyr et al., 2015). In general, simulations forced with transient climate from 1901 to 2009 (CRU TS3.10 monthly temperature and cloudiness, Harris, Jones, Osborn, & Lister, 2014; GPCC monthly precipitation, Schneider et al., 2014) were preceded by a 900-year spin-up without anthropogenic land use followed by a 120-year spin-up based on the fixed land-use pattern and recycling climate for 1901–1920. All simulation outputs were averaged over 1980-2009 to take out climate variability. For more details on the simulation protocol, see the Supplementary information in Gerten et al. (2020).

2.3 EFR calculation

As described in Gerten et al. (2020), EFRs were estimated based on the average monthly grid-cell discharge during the last 30 years of the model spin-up with potential natural (pristine) vegetation and climate input for 1951–1980, that is, without human land use and reservoirs and in the absence of water withdrawals. The reference river state thus constitutes a potential natural flow under current climatic conditions, albeit neglecting that some rivers have been anthropogenically modified for centuries. The variable monthly flow (VMF) method (Pastor, Ludwig, Biemans, Hoff, & Kabat, 2014) was used to determine monthly EFRs for each grid cell, which aim at sustaining river ecosystems at least

Hoff, & Kabat, 2014) was used to determine monthly EFRs for each grid cell, which aim at sustaining river ecosystems at least in a 'fair' status and were calculated based on flow regimedependent percentages of mean monthly flow (MMF) vs. mean annual flow (MAF) of the pristine discharge. We precautionarily took into account the considerable uncertainties in determining ecological water needs by increasing the published values of the VMF method by 15%, as proposed in Steffen et al. (2015): in lowflow months (MMF < $0.4 \times$ MAF), EFRs are estimated to be 75% of MMF; in intermediate-flow months (MMF > $0.4 \times MAF$ and $MMF \leq 0.8 \times MAF$) 60% of MMF and in high-flow months $(MMF > 0.8 \times MAF)$ 45% of MMF. The resulting EFR thresholds are used to constrain total water withdrawals in each month in the EFR scenario. Upon irrigation reduction, all crop stands receive proportionally less water. LPJmL simulated EFR estimates based on the VMF method have been successfully validated against local case study estimates in Jägermeyr et al. (2017).

2.4 Calorie calculation scheme

To implement dietary change scenarios, simulated yields from all CFTs were converted to crop and livestock calorie supply depending on livestock production and associated crop feed demand. As livestock production is not modeled in LPJmL, comprehensive post-processing calculations were performed based on data from the FAOSTAT database (FAOSTAT, 2017, 2018), among others on feed productivities and composition as well as dietary energy and protein content of different aliments. The applied calculation scheme summarized in the following and visualized in Figure 1, adapts and refines the scheme described in the Supplementary information of Gerten et al. (2020). For more details on single calculation steps and processing of external data, see Supplementary text and Figure S1.

2.4.1 Calculation of baseline livestock production

We derived the reference regional livestock production under current dietary habits (see reference diet in Figure 1) based on livestock production data from FAO Food Balance Sheets (FBS) for 2005 (FAOSTAT, 2017) and simulated pasture and roughage production from LPJmL. By dividing pasture and roughage production as simulated in the baseline scenario by total livestock production from FBS, roughage requirements per unit of average livestock product for 12 world regions were determined. To obtain current regional livestock production levels for each irrigation scenario, we divided the irrigation scenario's pasture and roughage production by these regional roughage requirements per livestock product. For countries not included in the FBS, we assigned the average roughage requirement of the world region the country belongs to.

2.4.2 Calculation of crop production for human consumption

To derive the regional crop production intended for direct human consumption (Crop Food), we removed CFT- and region-specific shares for 'other uses' (O) such as bioenergy and seed production as well as feed (F) from simulated edible plant production (P),

3



Fig. 1. Overview of scenarios and calorie calculation scheme used to convert simulated crop yields to regional calorie supply depending on the dietary scenario. Feed requirements per livestock product, other uses such as bioenergy and seed production as well as dietary energy and protein content for each CFT and 12 world regions were derived from FAO Food and Commodity Balance Sheets for 2005. Losses and waste amounts were taken from Gerten et al. (2020). Gray arrows visualize the effects of downscaling livestock production.

which was aggregated for the 13 CFTs and 12 world regions:

and the regional livestock production (LP):

$$Crop \ Food_{reg,cft} = P_{reg,cft} \times (1 - O_{reg,cft}) - F$$

$$F = FReq_{reg,cft} \times LP_{reg}$$

O is a product and country-dependent share of crop production allocated to these purposes according to the FBS utilization accounts, averaged for regions and CFTs by allocating FBS food items to CFTs, and assumed to be constant between different scenarios. F is calculated as the product of region-specific crop feed requirements per livestock product ($FReq_{reg,cft}$ = weight of each CFT needed to produce 1 kg of an average livestock product) $FReq_{reg,cft}$ was derived from FBS data for 2005 and adjusted to LPJmL production amounts (for details see Supplementary text). We also included food manufacture by-products used as feed (brans, molasses, oilseed cakes, allocated to the respective CFTs) obtained from the FAO Commodity Balances for 2005 (FAOSTAT, 2018). While trade flows within each of the 12 world regions are implicitly accounted for through regional

aggregation, our analysis of regional calorie supply potentials does not model trade flows between regions. Our estimates therefore reflect the domestic feed production potential.

2.4.3 Calculation of calorie and protein supply

To calculate livestock and crop fresh matter supply for human consumption from net production amounts, post-harvest/production losses, processing losses and food waste were subtracted using regionally averaged percentages (Gustavsson, Cederberg, Sonesson, van Otterdijk, & Meybeck, 2011; Jalava et al., 2016; Kummu et al., 2017). As FAO data exclude production losses and LPJmL crop yields were calibrated with these data, harvest losses are implicitly considered. The calorie and protein supply from livestock and crops was obtained by multiplying fresh matter supply with regional dietary energy and protein content derived from FBS food supply quantities in terms of energy, protein and mass units. While energy and protein contents of crops were specifically calculated for each CFT, the regional factors for livestock production represent an average over the sum of all livestock products with a composition depending on the respective regional production ratios. By averaging the livestock sector, we avoid complexity regarding co-products such as beef and milk, and the attribution of feed to individual livestock species.

2.5 Dietary changes

In order to investigate potential calorie supply changes resulting from dietary changes toward less livestock products, one reference diet with regional livestock production derived from FBS data (see above) and three dietary change scenarios were implemented. Building upon dietary change implementations published in Jalava et al. (2014), regional livestock production was incrementally downscaled so that protein share from livestock products is limited to 25% (DC25), 12.5% (DC12.5) and 0% (DC0) of total protein supply, respectively (Figure 1). In regions where livestock protein share is below 25%, livestock production is not altered in the DC25 scenario. While these dietary change scenarios refer to protein supply since livestock products primarily contribute to protein provision, results on total food supply changes refer to calories as this is the most relevant indicator for food provision. It is also important to note that the approach is production-based, thereby only indirectly reflecting changes in dietary habits. In contrast to the approach described in Gerten et al. (2020), the dietary change scenarios do not impact underlying land-use patterns. Calorie supply changes thus solely result from a changed calorie calculation scheme with (1) a decreased livestock production and supply and (2) reallocation of freed crop feed to direct human consumption (see gray arrows in Figure 1). Our analysis of regional calorie supply potentials allows us to locate these increases in crop calorie supply within the same region where feed demand is caused. Since food processing by-products are included in the CFT feed requirements, simple reallocation of livestock crop feed to direct human consumption would imply that these, in principle mostly human-edible products, are consumed by humans instead of livestock, which would require changes in consumption behavior. In this context, we assume that food processing by-products which are already currently partly being used for human food production can contribute to calorie supply (all molasses and brans; soybean-, groundnut-, coconut- and sesame seed-cakes) whereas the remaining oil seed cakes cannot (sunflower-, rape-, mustard-, cotton seed-

While the dietary changes would also reduce pasture feed requirements, potential indirect effects through abandonment of pasture areas and potential conversion to arable land or increases in uniquely grass-fed ruminants are not considered here. Since the livestock sector was averaged for each of the 12 regions, regional production ratios of different livestock products are preserved upon dietary change, which implies that all livestock products are reduced equally. Implementing more specific dietary scenarios such as a vegetarian diet was not possible due to the used input data: FAO does not provide feed composition per livestock product but only total feed quantity per crop type. Also, as the study focuses on reallocation of current crop production in terms of calories and proteins from feed to food, shifts to healthy and nutritious diets to increase food security especially in low-income countries, which would imply changes in cropping patterns, are not analyzed here.

2.6 Comparing the effects of EFR protection and dietary changes

To finally assess the combined effect of EFR protection and dietary changes on calorie supply for each world region and globally in an integrated framework, the dietary change scenarios were employed on the simulated reduced yield levels of the EFR scenario. Obtained calorie supply was then compared to calorie supply based on yield levels from the baseline scenario and current livestock production. Calorie supply thereby serves as an integrated 'response variable' for both EFR and diet change scenarios. We thus did not analyze water-saving potentials from dietary changes and its effect on EFR transgressions but – conversely – analyzed whether reallocating crop feed, both from rainfed and irrigated cropland, to direct human consumption could counteract calorie supply reductions upon EFR protection.

3. Results

3.1 Potential yield reductions upon EFR protection

According to the land-use dataset used, roughly 300 Mha of agricultural land were irrigated in 2005, corresponding to 18% of total arable land. If EFRs were to be protected globally following requirements derived from the VMF estimation method (EFR scenario), water withdrawals for irrigation on these areas would decrease from $2500 \text{ km}^3 \text{ year}^{-1}$ (baseline scenario, see Supplementary Table S2 for a comparison to other model-based estimates) to $1330 \text{ km}^3 \text{ year}^{-1}$ (averaged for 1980–2009 climate). This suggests that 47% of global irrigation water withdrawals are currently at the expense of EFR provision. These excess water uses occur primarily in irrigation hotspots of the northern hemisphere, for example, in South Asia, the Middle East and the Mediterranean (see Supplementary Figure S2). A reduction of available irrigation water would in turn impact agricultural yields: total global crop yields from irrigated areas would be reduced by 23.2% if EFRs were preserved globally. In other words, almost half of irrigation water withdrawals and almost a quarter of irrigated crop production rely on the transgressions of EFRs, at the expense of river ecosystem integrity and biodiversity. Given that 18% of cropland is irrigated, this translates into a total yield reduction (rainfed and irrigated together) of 5.2% upon EFR protection.



Fig. 2. Simulated impact of EFR protection on global and regional yields. (a) Percent decrease in dry matter (DM) crop yields (rainfed and irrigated) if EFRs were to be preserved globally. (b) Relative irrigated and total (irrigated and rainfed) yield reductions upon EFR protection aggregated for 12 world regions, ordered by magnitude. Top panel displays the share of yields from irrigated areas in the baseline scenario. All results are based on 2005 land use and 1980–2009 climate.

Strongest relative yield reductions are simulated in the Middle East and Central Asia. In some countries in these regions (e.g. Israel, Oman, Saudi Arabia, Uzbekistan and Turkmenistan), >50% of yields (dry matter, rainfed and irrigated together) rely on EFR transgressions (see Figure 2a). In terms of absolute yields, countries with strong EFR transgressions, like India, Pakistan, the United States and China, would be most severely affected by a protection of EFRs: 20.4% of total global yield reductions in the EFR scenario would occur in India alone, attaining 38.1% together with Pakistan (12%) and China (5.8%), and 11.5% of absolute reductions would be located in the United States. Against the backdrop that approximately one-third of globally undernourished people live in India and Pakistan alone (FAO et al., 2021), this result emphasizes the potential severe trade-off between food security and EFR protection in some regions.

Aggregating the results for the world regions emphasizes differences in the dependence of crop production on EFR transgressions. Relative reductions in yields due to EFR protection are highest in the Middle East with 28.7%, followed by South Asia (18.3%) and Australia and Oceania (13.2%) (Figure 2b). These regions are characterized by high shares of irrigated crop production (Figure 2b, top panel), due to low amounts of naturally available freshwater and/or irrigation-intensive crop growth such as rice, for example, in large parts of South Asia. Furthermore, irrigated drylands like in the western United States and southern Europe rely more on EFR transgressions than regions with less intensive agriculture, such as sub-Saharan Africa. In East Asia, yield reductions are low despite high irrigation shares, as EFRs are simulated to be maintained in many parts due to the rain-laden subtropical monsoon climate which cover large parts of south-east China.

3.2 Potential calorie supply increases through dietary changes

To assess whether the (hypothetical) negative effects of EFR provision on regional and global calorie supply could potentially be

buffered or compensated for by dietary changes, the simulated crop yields from the baseline scenario were converted to calorie supply depending on livestock production levels. By consecutively down-scaling the protein supply from livestock products from the current levels to a maximum of 25% (DC25), 12.5% (DC12.5) and 0% (DC0) of total protein supply in each world region, the associated shifts in crop use from feed to food could increase global calorie supply as the nutritionally inefficient conversion from feed to livestock product is circumvented (Figure 3a, 'GLOBAL'). Even without considering the possible additional use of freed pasture areas for crop growth, global calorie supply in the DC25 scenario is calculated to increase by 4.2%. The additional calorie supply from crop feed shifted to human consumption thus overcompensates the mean global reductions of livestock calories (-24.6%). A mean reduction of livestock calories by 59.6% in the DC12.5 scenario leads to an overall calorie supply increase of 10.7% and in the scenario without livestock production (DC0), calorie supply increases by 19.0% at the global level (for a sensitivity analysis regarding the edibility of feed for humans, see Supplementary Table S1).

While the DC25 scenario entails a 23% increase of calorie supply in Western Europe, the DC0 scenario even indicates a calorie supply gain of about 48% both in Western Europe and North America (see Figure 3a; for regional percentages of calorie supply change for all three scenarios see Supplementary Table S1). In terms of absolute calorie supply, 50% of the increases in the DC0 scenario occur in Western Europe and North America alone, attaining 75% together with Eastern Europe and Central Asia and East Asia. The high effectivity of dietary changes calculated for these regions can be attributed to (1) high initial livestock calorie share and (2) high crop feed shares, that is, crop feed divided by total feed amount as calculated with the applied calorie calculation scheme (see Figure 4). In other words, for Western Europe, North America and Eastern Europe and Central Asia, assumed crop feed shares are relatively high and



Fig. 3. Impacts of dietary changes on calorie supply in the context of EFR protection, aggregated to world regions. (a) Relative change of calorie supply in comparison with the baseline scenario (current irrigation practices, livestock production as derived from FAO data for 2005), resulting from EFR protection (red bars) and dietary change scenarios with gradually reduced livestock production (light gray bars) to a maximum of 25% (DC25), 12.5% (DC12.5) and 0% (DC0) of total protein supply. Regions are ordered by their absolute calorie supply (dark gray bars in the bottom). Asterisks (*) mark regions with negative effects of dietary changes. (b) Calorie supply changes resulting from combined effects of EFR protection and the three dietary change scenarios.

dietary change therefore frees up more cropland for food use. Conversely, the crop feed shares in South Asia and Australia and Oceania are very low, which leads to a negative effect of dietary change on total calorie supply: if livestock production is reduced, only small amounts of crop feed are reallocated to direct human consumption. Here, calculated calorie supply increases from plant-based products are smaller than animal calorie supply reductions. This is in line with observational data on the high share of grass-fed and/or by-product-fed livestock in these regions (Herrero et al., 2013). Also, the DC25 scenario does not change calorie supply at all in South Asia, Southeast Asia and Southern Africa given that livestock protein share in these regions is already below 25% (see Figure 4).

3.3 Potential compensation of EFR protection effects through dietary changes

Our further analysis suggests that comparatively modest dietary changes (DC25 scenario) could partly compensate for calorie

supply reductions resulting from EFR protection (see Figure 3a). Stronger livestock reductions as in the DC12.5 and DC0 scenarios could even largely overcompensate the effects of EFR provision on the global level. Globally, EFR and DC effects would cancel each other out if animal protein share was capped at 19.4%, corresponding to a mean calorie share from animal products of 10.4% and a global mean reduction of the livestock calorie share by 42%. While this implies drastic large-scale dietary changes, such limited intake of livestock products could also benefit health and individual well-being (Godfray et al., 2018; Tilman & Clark, 2014; Willett et al., 2019).

Analyzing regional potentials of dietary changes to compensate for EFR protection effects results in a heterogeneous picture, suggesting that regions with strong reliance on EFR transgressions do not necessarily show high dietary change potentials and vice versa (Figure 3b). Limiting the animal protein share to 25% (DC25) suffices to at least compensate calorie reductions from EFR provision in Central America, Eastern Europe and Central Asia, North America and Western Europe (Figure 3b, top map). Kcal change (DC0)Livestock kcal shareLivestock protein shareCrop kcal used for feedCrop feed shareEast Asia8.421.834.827.741.3South Asia-3.711.522.510.14.5North America48.219.729.340.741.4

South Asia	-9.1	11.5	22.3	10.1	4.5
North America	48.2	19.7	29.3	40.7	41.4
Southeast Asia	13.3	5.3	21.3	15.8	14.7
East. Eur. & Central Asia	38.0	20.9	35.6	39.1	37.4
Western Europe	48.3	36.3	53.5	54.8	43.5
South America	13.2	19.2	28.3	25.6	12.8
Southern Africa	17.2	8.3	22.6	21.2	11.6
Middle East	20.9	18.1	36.0	31.8	21.9
Central America	20.3	17.7	44.6	36.8	22.9
North Africa	9.8	21.6	42.8	28.4	10.8
Australia & Oceania -14	1.0	37.1	47.4	27.0	4.4
GLOBAL	19.0	18.0	32.6	29.6	23.0

Fig. 4. Domestic production-based estimates of regional characteristics (in %), which underlie the regional dietary change potential, that is, the change in total calorie supply if edible crop feed was used for human consumption (DC0 scenario, gray bars). Displayed are livestock production contributing to total calorie supply and total protein supply as well as the share of crop calorie production allocated to feed and the share of crop feed in total feed amount (roughage and grass included) for a regionally specific average livestock product. All factors are based on FAO-derived input data for 2005 and the calorie calculation scheme as described in the Methods.

A further reduction of the animal protein share to 12.5% in each region (DC12.5) additionally leads to a compensation of EFR protection induced calorie supply reductions in East Asia, Southeast Asia, South America and Southern Africa (Figure 3b, middle map). Dietary changes are not sufficient to compensate for calorie supply reductions in the Middle East and North Africa, and in Australia and Oceania and South Asia, dietary changes might even be counterproductive to sustain regional food supply due to low crop feed shares (Figure 3b, bottom map). As such, the results suggest that in some regions with high reliance of food production on EFR transgressions, which are partly also severely impacted by food insecurity (i.e. South Asia), dietary changes do not seem appropriate to solve the trade-off between EFR protection and food supply. At the same time, it is important to note that in these regions (1) disproportional land use through pasturefed animals and resulting possibilities to convert pastures to arable land or to maintain pasture-fed livestock production were not considered in this study, and (2) international trade among regions could compensate for such deficits by balancing limited calorie production in these regions.

4. Discussion

We show that, at the global level, dietary changes up to a scenario with no livestock production could clearly overcompensate calorie supply reductions that would occur if irrigation was limited to maintain EFRs worldwide. Thus, current calorie production inefficiencies due to feeding animals with human edible crops (Shepon, Eshel, Noor, & Milo, 2018) seem to be far more pronounced than current crop production amounts relying on transgressions of environmental flows. In other words, diet changes could compensate for potential calorie supply losses, not directly due to lowered water demands, but due to calorie and protein losses inherent in livestock production. In the following we compare results on effects of EFR protection and dietary changes to literature, discuss limitations and embed our findings in a broader context of food supply within the planetary boundary for freshwater use.

The simulated 47% reduction of irrigation water withdrawals upon global EFR protection is in good agreement with earlier estimates: 41% in Jägermeyr et al. (2017) (the same model but average based on four hydrological EFR methods), 40% in Rosa et al. (2018) (different model but the same EFR calculation method), 52% in Rosa, Chiarelli, Tu, Rulli, and D'Odorico (2019) (different model, more conservative EFR estimate based on annual flow, depletion of groundwater stocks included) and 39% in Droppers et al. (2020) (different model, the same EFR method but without precautionary + 15%, additional EFRs for groundwater baseflow). Besides uncertainties in modeled discharge, reservoir management and irrigation water withdrawals resulting, for example, from different land use (Puy, Borgonovo, Lo Piano, Levin, & Saltelli, 2021) and climate data (Biemans et al., 2009), these differences can also be explained by more or less conservative EFR estimates: hydrological EFR methods such as the one applied here represent rough estimates and need local refinements using more holistic methods (Pastor et al., 2014). Additionally, their definition also comprises a normative component regarding the 'best compromise' between human and environmental water needs. Furthermore, not considering non-renewable groundwater abstractions and groundwater depletion, might lead to an overestimation of water withdrawals from surface waters and, thus, EFR transgressions in regions with high irrigation volumes from groundwater such as Northern India, Western United States and Pakistan (see Gerten et al., 2020). Fossil groundwater abstractions can however be regarded as unsustainable which justifies not to take them into account in the 'sustainable' EFR scenario which should neither invoke EFR transgression nor depletion of fossil aquifers. Also, groundwater abstractions translate into decreased river streamflow even before substantial groundwater depletion

occurs (de Graaf et al., 2019) and thus significantly contribute to EFR transgressions (Droppers et al., 2020). This might also explain why our estimate of irrigation water withdrawals relying on EFR transgressions is in good agreement with other estimates which explicitly consider groundwater depletion (see above).

Due to the more precautionary estimate of EFRs used in this study in line with the current definition of the planetary boundary for freshwater use (Steffen et al., 2015), the total yield reduction (5.2%) resulting from EFR protection is slightly higher than the 4.6% found by Jägermeyr et al. (2017).

Regarding the potential calorie supply increases through direct use of crops grown for animal feed, our estimate (19% in the DC0 scenario) is lower than previous estimates (Cassidy et al., 2013; Foley et al., 2011). These estimates include more by-products used as feed, for example, from bioenergy production and/or assume that more by-products could be directly consumed as human food. In accordance with this, the share of calorie production used for feed calculated in this study is lower than in other global estimates (see Supplementary Table S2 for an evaluation of global estimates of simulated key variables against independent datasets). Also, global crop production in LPJmL is generally slightly underestimated (Jägermeyr et al., 2017). Due to the adjustment of feed requirements to LPJmL production amounts, this results in lower feed requirements per livestock product. Therefore, the here presented estimates of the potentials of dietary changes to increase calorie supply can be seen as conservative.

Our results also show that potential calorie supply increases through dietary changes are geographically concentrated in regions with high percentages of concentrate feed and calorie supply from animal products: 75% of absolute calorie supply potentials in the DC0 scenario are located in North America, Western Europe, Eastern Europe and Central Asia and East Asia. This is in good agreement with West et al. (2014) who found that shifting crop feed to direct human consumption in the United States, Western Europe and China could account for 54% of the global 'diet gap'. Dietary changes in these regions can thus be regarded as a major leverage point to sustainably increase food supply within limited resources, among them water (West et al., 2014).

The uneven distribution of effects does not only underpin (1) the disproportionally high responsibilities of countries with resource-intensive diets and (2) the importance of trade as a vehicle to provide water-scarce regions with food from regions with high calorie supply increases upon dietary changes. It also points to (3) the need for regionally adapted solutions and a broad spectrum of measures addressing both production and demand. Such measures comprise among others a sustainable intensification of agriculture, for example, through better water management, increased livestock water productivity or sustainable expansion of irrigation particularly in low-income countries (Heinke et al., 2020; Jägermeyr et al., 2017; Rosa et al., 2018), as well as reductions of food losses and waste (Gerten et al., 2020; Kummu et al., 2012; Springmann et al., 2018). Given the strong alteration of streamflow through dams and reservoirs in many basins (Grill et al., 2015), removal of dams or optimized operation rules for reservoir outflow could also help to sustain EFRs by minimizing flow regime alteration (Yin, Yang, & Petts, 2011). Such technical improvements might provide near- or mid-term solutions in addition to rather long-lasting societal transformations as required for example for large-scale dietary changes. While we looked at one option to ease trade-offs between ecosystem water needs and food security, the combined effects of several measures could be simulated integratively in further research.

Regarding livestock production, the heterogeneity in regional calorie supply improvements due to dietary changes also suggests that reduced consumption of livestock products does not necessarily increase calorie supply within limited freshwater resources. While allocation of human-edible crop feed to livestock production usually represents a net loss of human available calories and causes a competition between feed and food for valuable natural resources such as land and water (Foley et al., 2011), livestock uniquely fed by food processing by-products, crop residues, food waste and grass could avoid this competition (Nonhebel, 2004; Schader et al., 2015). This is in line with the simulated low potentials of dietary changes to increase food supply in regions with low crop feed shares, such as South Asia and Australia and Oceania. Additionally, livestock production can contribute valuable proteins, vitamins and micronutrients especially in low-income countries and may generate income at the household level in rural regions and thereby support livelihoods (Garnett, 2013; Mottet et al., 2017). Globally limiting livestock production to marginal pastures and leftovers could thus serve both food security and freshwater ecosystem protection by decoupling livestock production from arable land and its irrigation (Van Zanten et al., 2018). We here assumed that all livestock production systems and products are reduced equally in the DC scenarios, concentrating on effects from reallocation of crop-feed. Further studies could however represent dietary change scenarios, which specifically decrease crop-fed livestock systems while maintaining or increasing 'crop-feed-free' systems. This would necessitate an extended calorie calculation scheme representing feed flows for distinct livestock production systems and products. A first rough estimate based on calculations from Van Zanten, Meerburg, Bikker, Herrero, and de Boer (2016) indicates that extensive grazing on all global pasture areas (2961 Mha according to our land-use input) could provide 2.5×10^{14} kcal and 2.1×10^{10} kg proteins from ruminant meat and milk per year (see additional text and calculations in Supplementary Table S8). This could increase the calorie supply in the DC0 scenario (=0% crop feed for livestock) by an additional 4%, from 19 to 23% in comparison with the baseline scenario with current livestock production. Studies, which explicitly examined potential livestock production without relying on crop feed (Röös et al., 2017; Schader et al., 2015; Van Zanten et al., 2016, 2018), estimate an even higher potential global production if not only grass but also crop residues, food waste and/or by-products were extensively exploited as feed in future (see Supplementary Table S9 and explanatory Supplementary text). In light of a growing world population, such a 'sustainable boundary for livestock production' would imply a reduction of per capita livestock production and consumption at the global level with strong reductions in most high-income regions, but still allow for a restricted growth in parts of Africa and Asia (Van Zanten et al., 2018). From both a food security and EFR protection perspective, such a 'crop-feed-free' livestock production could be preferable over the stylized diet without livestock products as implemented in the DC0 scenario. On the other hand, abandonment of some pasture areas currently used for animal grazing as implicitly assumed in our DC scenarios could also significantly contribute to attaining other sustainability targets such as biodiversity protection and climate change mitigation, for example, through reforestation of pasture areas, which replaced natural forest ecosystems in the past, or by rewilding overgrazed areas (Hayek, Harwatt, Ripple, & Mueller, 2021; Kemppinen et al., 2020).

Both the EFR scenario and the DC scenarios do not include secondary effects such as changes in land use and trade. The scenarios are thus not representing pathways but explicitly isolate and compare the effect of EFR provision and dietary changes on calorie supply. Thereby, they offer an opportunity to study their individual effects based on spatially detailed, process-based calculations of yields and water fluxes. To elucidate the complex interactions, tradeoffs and synergies of dietary choices as well as water regulation policies with land-use patterns and trade flows, simulations with integrated assessment models could provide additional insights (Bonsch et al., 2015; Kim et al., 2016; Pastor et al., 2019; Weindl et al., 2017). Also, including trade in further studies would likely shift some of the calories supply increases upon DC from countries with high livestock production, such as Western Europe, to feed producing countries, for example, the Americas for soy (Wang et al., 2018). Furthermore, it is likely and from a nutritional perspective desirable that reductions in animal product consumption are accompanied by increases in fruit, vegetable, nut, oil and legume consumption (Tilman & Clark, 2014; Willett et al., 2019). Regarding protein supply, the energy share contributed by proteins would only be slightly reduced through the implemented dietary change scenarios and would still remain within healthy limits at a global level (see Supplementary Table S3). Nevertheless, increases in protein-rich plants are important to sustain or increase protein intake upon dietary changes in low-income countries. While these plant-based commodities are in general substantially less resource-intensive than animal products, they may consume more natural resources than human edible animal feed such as feed grains and oil seeds (Poore & Nemecek, 2018). Transitioning to healthy, primarily plant-based diets for all may thus pose more challenges with regard to EFR compatible food supply than simulated here, in particular for the cultivation of vegetables and fruits, which are often irrigated. It is also important to note that dietary changes, unaccompanied by EFR protection policies, might not directly lead to reductions in EFR transgressions.

This study is designed to analyze combined effects of ambitious measures targeting different levels of the food system, thus contributing to the broader discussion on food supply within planetary boundaries (Gerten et al., 2020). It confirms that dietary change from animal to plant-based products is a powerful measure to increase calorie supply within environmental constraints such as EFR protection. However, current trends in water withdrawals and diets (Bodirsky et al., 2020) point to the opposite direction. With a growing population and projected increases in livestock product consumption especially in developing nations, water and land requirements for agriculture are expected to further grow in the future (Tilman, Balzer, Hill, & Befort, 2011). Additionally, climate change and biomass-based climate change mitigation strategies will put both agriculture and natural resources under increasing pressures (Bonsch et al., 2016; Gosling & Arnell, 2016; Schewe et al., 2014). Despite large implementation obstacles (Arthington et al., 2018; Eker, Reese, & Obersteiner, 2019), both EFR protection and dietary changes seem indispensable in a broader context of the twin challenge of reaching food security within environmental limits (Jägermeyr, 2020) and need more political and social attention if ecological and social sustainability targets are to be met jointly.

Supplementary material. The supplementary material for this article can be found at https://doi.org/10.1017/sus.2022.12

Acknowledgments. We thank Vera Heck for providing analysis scripts.

Author contributions. JB, FS and DG designed the study. JB analyzed the data and wrote the manuscript draft, with support from FS and DG. BLB contributed to specific parts of the study design and MJ to specific parts of the data analysis. All co-authors commented on and revised the manuscript.

Financial support. This work was supported by the European Union's Horizon 2020 research and innovation program (JB, grant agreement No. 869192), the German Federal Ministry for Environment, Nature Conservation, Building and Nuclear Safety (JB and FS, IKI project No. 16_II_148_Global_ A_IMPACT) and the MVTT (MJ). FS additionally acknowledges funding by the CE-Land+ project of the German Research Foundation's priority program SPP 1689 'Climate Engineering – Risks, Challenges and Opportunities?'.

Conflict of interest. The authors declare they have no conflict of interest.

Data and code availability. Data supporting the results of this study are provided in the supplementary material. Model code and analysis scripts are available from the corresponding author on request.

References

- Arthington, A. H., Bhaduri, A., Bunn, S. E., Jackson, S. E., Tharme, R. E., Tickner, D., Young, B., Acreman, M., Baker, N., Capon, S., Horne, A. C., Kendy, E., McClain, M. E., Poff, N. L. R., Richter, B. D., & Ward, S. (2018). The Brisbane declaration and global action agenda on environmental flows (2018). *Frontiers in Environmental Science*, 6(45), 1–15. https://doi. org/10.3389/fenvs.2018.00045.
- Berners-Lee, M., Kennelly, C., Watson, R., & Hewitt, C. N. (2018). Current global food production is sufficient to meet human nutritional needs in 2050 provided there is radical societal adaptation. *Elementa: Science of the Anthropocene*, 6(52), 1–14. https://doi.org/10.1525/elementa.310.
- Biemans, H., Haddeland, I., Kabat, P., Ludwig, F., Hutjes, R. W. A., Heinke, J., von Bloh, W., & Gerten, D. (2011). Impact of reservoirs on river discharge and irrigation water supply during the 20th century. *Water Resources Research*, 47(3), 1–15. https://doi.org/10.1029/2009WR008929.
- Biemans, H., Hutjes, R. W. A., Kabat, P., Strengers, B. J., Gerten, D., & Rost, S. (2009). Effects of precipitation uncertainty on discharge calculations for main river basins. *Journal of Hydrometeorology*, 10(4), 1011–1025. https:// doi.org/10.1175/2008JHM1067.1.
- Bodirsky, B. L., Dietrich, J. P., Martinelli, E., Stenstad, A., Pradhan, P., Gabrysch, S., Mishra, A., Weindl, I., Le Mouël, C., Rolinski, S., Baumstark, L., Wang, X., Waid, J. L., Lotze-Campen, H., & Popp, A. (2020). The ongoing nutrition transition thwarts long-term targets for food security, public health and environmental protection. *Scientific Reports*, 10 (19778), 1–14. https://doi.org/10.1038/s41598-020-75213-3.
- Bonsch, M., Humpenöder, F., Popp, A., Bodirsky, B., Dietrich, J. P., Rolinski, S., Biewald, A., Lotze-Campen, H., Weindl, I., Gerten, D., & Stevanovic, M. (2016). Trade-offs between land and water requirements for large-scale bioenergy production. *GCB Bioenergy*, 8, 11–24. https://doi.org/10.1111/gcbb. 12226.
- Bonsch, M., Popp, A., Biewald, A., Rolinski, S., Schmitz, C., Weindl, I., Stevanovic, M., Högner, K., Heinke, J., Ostberg, S., Dietrich, J. P., Bodirsky, B., Lotze-Campen, H., & Humpenöder, F. (2015). Environmental flow provision: Implications for agricultural water and land-use at the global scale. *Global Environmental Change*, 30, 113–132. https://doi.org/10.1016/j.gloenvcha.2014. 10.015.
- Campbell, B. M., Beare, D. J., Bennett, E. M., Hall-Spencer, J. M., Ingram, J. S. I., Jaramillo, F., Ortiz, R., Ramankutty, N., Sayer, J. A., & Shindell, D. (2017). Agriculture production as a major driver of the earth system exceeding planetary boundaries. *Ecology and Society*, 22(4), 1–11. https://doi.org/10. 5751/ES-09595-220408.
- Cassidy, E. S., West, P. C., Gerber, J. S., & Foley, J. A. (2013). Redefining agricultural yields: From tonnes to people nourished per hectare. *Environmental Research Letters*, 8, 1–8. https://doi.org/10.1088/1748-9326/ 8/3/034015.
- Davis, K. F., Rulli, M. C., Seveso, A., & D'Odorico, P. (2017). Increased food production and reduced water use through optimized crop distribution. *Nature Geoscience*, 10, 919–924. https://doi.org/10.1038/s41561-017-0004-5.

- de Graaf, I. E. M., Gleeson, T., (Rens) van Beek, L. P. H., Sutanudjaja, E. H., Edwin, H., & Bierkens, M. F. P. (2019). Environmental flow limits to global groundwater pumping. *Nature*, 574(7776), 90–94. http://dx.doi.org/10. 1038/s41586-019-1594-4.
- Döll, P., Hoffmann-Dobrev, H., Portmann, F. T., Siebert, S., Eicker, A., Rodell, M., Strassberg, G., & Scanlon, B. R. (2012). Impact of water withdrawals from groundwater and surface water on continental water storage variations. *Journal* of *Geodynamics*, 59, 143–156. https://doi.org/10.1016/j.jog.2011.05.001.
- Döll, P., Müller Schmied, H., Schuh, C., Portmann, F. T., & Eicker, A. (2014). Global-scale assessment of groundwater depletion and related groundwater abstractions: Combining hydrological modeling with information from well observations and GRACE satellites. *Water Resources Research*, 50(7), 5698– 5720. https://doi.org/10.1002/2014WR015595.
- Droppers, B., Franssen, W. H. P., van Vliet, M. T. H., Nijssen, B., & Ludwig, F. (2020). Simulating human impacts on global water resources using VIC-5. *Geoscientific Model Development*, 13(10), 5029–5052. https://doi.org/10. 5194/gmd-13-5029-2020.
- Eker, S., Reese, G., & Obersteiner, M. (2019). Modelling the drivers of a widespread shift to sustainable diets. *Nature Sustainability*, 2(8), 725–735. https://doi.org/10.1038/s41893-019-0331-1.
- Fader, M., Rost, S., Müller, C., Bondeau, A., & Gerten, D. (2010). Virtual water content of temperate cereals and maize: Present and potential future patterns. *Journal of Hydrology*, 384(3–4), 218–231. https://doi.org/10.1016/j. jhydrol.2009.12.011.
- FAO. (2020). AQUASTAT Core Database. Food and Agricultural Organization of the United Nations. Website accessed on 22.02.2020. http://www.fao.org/ aquastat/en/overview/.
- FAO, IFAD, UNICEF, WFP, & WHO. (2021). The state of food security and nutrition in the world 2021. Transforming food systems for food security, improved nutrition and affordable healthy diets for all. Rome: FAO. https://doi.org/10.4060/cb4474en.
- FAOSTAT. (2017). FAOSTAT database: Food balance sheets. Food and agriculture organization of the united nations (FAO). Rome: FAO. http://www.fao. org/faostat/en/#data/FBSH.
- FAOSTAT. (2018). FAOSTAT database: Commodity balances. Food and agriculture organization of the united nations (FAO). Rome: FAO. http://www. fao.org/faostat/en/#data/BC.
- FAOSTAT. (2020). FAOSTAT database: Crop production. Food and agriculture organization of the united nations (FAO). Rome: FAO. http://www.fao.org/faostat/en/#data/QC.
- Flörke, M., Kynast, E., Bärlund, I., Eisner, S., Wimmer, F., & Alcamo, J. (2013). Domestic and industrial water uses of the past 60 years as a mirror of socioeconomic development: A global simulation study. *Global Environmental Change*, 23(1), 144–156. https://doi.org/10.1016/j.gloenvcha.2012.10.018.
- Foley, J. A., Ramankutty, N., Brauman, K. A., Cassidy, E. S., Gerber, J. S., Johnston, M., Mueller, N. D., O'Connell, C., Ray, D. K., West, P. C., Balzer, C., Bennett, E. M., Carpenter, S. R., Hill, J., Monfreda, C., Polasky, S., Rockström, J., Sheehan, J., Siebert, S., ... Zaks, D. P. M. (2011). Solutions for a cultivated planet. *Nature*, 478(7369), 337–342. https://doi.org/10.1038/nature10452.
- Garnett, T. (2013). Food sustainability: Problems, perspectives and solutions. Proceedings of the Nutrition Society, 72(1), 29–39. https://doi.org/10.1017/ S0029665112002947.
- Gerten, D., Heck, V., Jägermeyr, J., Bodirsky, B. L., Fetzer, I., Jalava, M., Kummu, M., Lucht, W., Rockström, J., Schaphoff, S., & Schellnhuber, H. J. (2020). Feeding ten billion people is possible within four terrestrial planetary boundaries. *Nature Sustainability*, 3, 200–208. https://doi.org/10.1038/s41893-019-0465-1.
- Gerten, D., & Kummu, M. (2021). Feeding the world in a narrowing safe operating space. One Earth, 4(9), 1193–1196. https://doi.org/10.1016/j.oneear. 2021.08.020.
- Gleeson, T., Wang-Erlandsson, L., Zipper, S. C., Porkka, M., Jaramillo, F., Gerten, D., Fetzer, I., Cornell, S. E., Piemontese, L., Gordon, L. J., Rockström, J., Oki, T., Sivapalan, M., Wada, Y., Brauman, K. A., Flörke, M., Bierkens, M. F. P., Lehner, B., Keys, P., ... Famiglietti, J. S. (2020). The water planetary boundary: Interrogation and revision. *One Earth*, *2*, 223–234. https://doi.org/10.1016/j.oneear.2020.02.009.
- Godfray, H. C. J., Aveyard, P., Garnett, T., Hall, J. W., Key, T. J., Lorimer, J., Pierrehumbert, R. T., Scarborough, P., Springmann, M., & Jebb, S. A.

(2018). Meat consumption, health, and the environment. *Science* (*New York, N.Y.*), 361(6399), 1–10. https://doi.org/10.1126/science. aam5324.

- Godfray, H. C. J., Beddington, J. R., Crute, I. R., Haddad, L., Lawrence, D., Muir, J. F., Pretty, J., Robinson, S., Thomas, S. M., & Toulmin, C. (2010). Food security: The challenge of feeding 9 billion people. *Science* (*New York, N.Y.*), 327(5967), 812–818. https://doi.org/10.1126/science. 1185383.
- Gosling, S. N., & Arnell, N. W. (2016). A global assessment of the impact of climate change on water scarcity. *Climatic Change*, 134, 371–385. https:// doi.org/10.1007/s10584-013-0853-x.
- Grill, G., Lehner, B., Lumsdon, A. E., Macdonald, G. K., Zarfl, C., & Reidy Liermann, C. (2015). An index-based framework for assessing patterns and trends in river fragmentation and flow regulation by global dams at multiple scales. *Environmental Research Letters*, 10(1), 1–15. https://doi. org/10.1088/1748-9326/10/1/015001.
- Gustavsson, J., Cederberg, C., Sonesson, U., van Otterdijk, R., & Meybeck, A. (2011). Global food losses and food waste: Extent, causes and prevention. FAO.
- Harris, I., Jones, P. D., Osborn, T. J., & Lister, D. H. (2014). Updated highresolution grids of monthly climatic observations – The CRU TS3.10 dataset. *International Journal of Climatology*, 34, 623–642. https://doi.org/10. 1002/joc.3711.
- Hayek, M. N., Harwatt, H., Ripple, W. J., & Mueller, N. D. (2021). The carbon opportunity cost of animal-sourced food production on land. *Nature Sustainability*, 4(1), 21–24. https://doi.org/10.1038/s41893-020-00603-4.
- Heinke, J., Lannerstad, M., Gerten, D., Havlik, P., Herrero, M., Notenbaert, A., Hoff, H., & Müller, C. (2020). Water use in global livestock production-opportunities and constraints for increasing water productivity. *Water Resources Research*, 56(12), 1–16. https://doi.org/10.1029/ 2019WR026995.
- Herrero, M., Havlík, P., Valin, H., Notenbaert, A., Rufino, M. C., Thornton, P. K., Blümmel, M., Weiss, F., Grace, D., & Obersteiner, M. (2013). Biomass use, production, feed efficiencies, and greenhouse gas emissions from global livestock systems. *Proceedings of the National Academy of Sciences of the United States of America*, 110(52), 20888–20893. https://doi.org/10.1073/pnas.1308149110.
- Hoekstra, A. Y. (2012). The hidden water resource use behind meat and dairy. *Animal Frontiers*, 2(2), 3–8. https://doi.org/10.2527/af.2012-0038.
- Hogeboom, R. J., Bruin, D., Schyns, J. F., Krol, M. S., & Hoekstra, A. Y. (2020). Capping human water footprints in the world's river basins. *Earth's Future*, 8(2), 1–14. https://doi.org/10.1029/2019ef001363.
- Jägermeyr, J. (2020). Agriculture's historic twin-challenge toward sustainable water use and food supply for all. Frontiers in Sustainable Food Systems, 4(35), 1–16. https://doi.org/10.3389/fsufs.2020.00035.
- Jägermeyr, J., Gerten, D., Heinke, J., Schaphoff, S., Kummu, M., & Lucht, W. (2015). Water savings potentials of irrigation systems: Global simulation of processes and linkages. *Hydrology and Earth System Sciences*, 19(7), 3073– 3091. https://doi.org/10.5194/hess-19-3073-2015.
- Jägermeyr, J., Pastor, A., Biemans, H., & Gerten, D. (2017). Reconciling irrigated food production with environmental flows for Sustainable Development Goals implementation. *Nature Communications*, 8(15900), 1–9. https://doi.org/10.1038/ncomms15900.
- Jalava, M., Guillaume, J. H. A., Kummu, M., Porkka, M., Siebert, S., & Varis, O. (2016). Diet change and food loss reduction: What is their combined impact on global water use and scarcity? *Earth's Future*, 4(3), 62–78. https://doi.org/ 10.1002/2015EF000327.
- Jalava, M., Kummu, M., Porkka, M., Siebert, S., & Varis, O. (2014). Diet change – A solution to reduce water use? *Environmental Research Letters*, 9, 1–14. https://doi.org/10.1088/1748-9326/9/7/074016.
- Kemppinen, K. M. S., Collins, P. M., Hole, D. G., Wolf, C., Ripple, W. J., & Gerber, L. R. (2020). Global reforestation and biodiversity conservation. *Conservation Biology*, 34(5), 1221–1228. https://doi.org/10.1111/cobi.13478.
- Kim, S. H., Hejazi, M., Liu, L., Calvin, K., Clarke, L., Edmonds, J., Kyle, P., Patel, P., Wise, M., & Davies, E. (2016). Balancing global water availability and use at basin scale in an integrated assessment model. *Climatic Change*, *136*, 217–231. https://doi.org/10.1007/s10584-016-1604-6.
- Kummu, M., de Moel, H., Porkka, M., Siebert, S., Varis, O., & Ward, P. J. (2012). Lost food, wasted resources: Global food supply chain losses and

their impacts on freshwater, cropland, and fertiliser use. *Science of the Total Environment*, 438, 477–489. https://doi.org/10.1016/j.scitotenv.2012.08.092.

- Kummu, M., Fader, M., Gerten, D., Guillaume, J. H., Jalava, M., Jägermeyr, J., Pfister, S., Porkka, M., Siebert, S., & Varis, O. (2017). Bringing it all together: Linking measures to secure nations' food supply. *Current Opinion in Environmental Sustainability*, 29, 98–117. https://doi.org/10. 1016/j.cosust.2018.01.006.
- Lehner, B., Liermann, C. R., Revenga, C., Vörösmarty, C., Fekete, B., Crouzet, P., Döll, P., Endejan, M., Frenken, K., Magome, J., Nilsson, C., Robertson, J. C., Rödel, R., Sindorf, N., & Wisser, D. (2011). High-resolution mapping of the world's reservoirs and dams for sustainable river-flow management. *Frontiers in Ecology and the Environment*, 9(9), 494–502. https://doi.org/ 10.1890/100125.
- Liu, J., Hertel, T. W., Lammers, R. B., Prusevich, A., Baldos, U. L. C., Grogan, D. S., & Frolking, S. (2017). Achieving sustainable irrigation water withdrawals: Global impacts on food security and land use. *Environmental Research Letters*, 12, 1–11. https://doi.org/10.1088/1748-9326/aa88db.
- Mottet, A., de Haan, C., Falcucci, A., Tempio, G., Opio, C., & Gerber, P. (2017). Livestock: On our plates or eating at our table? A new analysis of the feed/food debate. *Global Food Security*, 14, 1–8. https://doi.org/10. 1016/j.gfs.2017.01.001.
- Nonhebel, S. (2004). On resource use in food production systems: The value of livestock as 'rest-stream upgrading system'. *Ecological Economics*, 48(2), 221–230. https://doi.org/10.1016/j.ecolecon.2003.09.013.
- Pang, A., Li, C., Sun, T., Yang, W., & Yang, Z. (2018). Trade-off analysis to determine environmental flows in a highly regulated watershed. *Scientific Reports*, 8, 1–11. https://doi.org/10.1038/s41598-018-32126-6.
- Pastor, A. V., Ludwig, F., Biemans, H., Hoff, H., & Kabat, P. (2014). Accounting for environmental flow requirements in global water assessments. *Hydrology and Earth System Sciences*, 18(12), 5041–5059. https:// doi.org/10.5194/hess-18-5041-2014.
- Pastor, A. V., Palazzo, A., Havlik, P., Biemans, H., Wada, Y., Obersteiner, M., Kabat, P., & Ludwig, F. (2019). The global nexus of food-trade-water sustaining environmental flows by 2050. *Nature Sustainability*, 2(6), 499–507. https://doi.org/10.1038/s41893-019-0287-1.
- Poore, J., & Nemecek, T. (2018). Reducing food's environmental impacts through producers and consumers. *Science (New York, N.Y.)*, 360(6392), 987–992. https://doi.org/10.1126/science.aaq0216.
- Portmann, F. T., Siebert, S., & Döll, P. (2010). MIRCA2000-Global monthly irrigated and rainfed crop areas around the year 2000: A new highresolution data set for agricultural and hydrological modeling. *Global Biogeochemical Cycles*, 24(1), 1–24. https://doi.org/10.1029/2008gb003435.
- Puy, A., Borgonovo, E., Lo Piano, S., Levin, S. A., & Saltelli, A. (2021). Irrigated areas drive irrigation water withdrawals. *Nature Communications*, 12(1), 4525. https://doi.org/10.1038/s41467-021-24508-8.
- Raworth, K. (2012). A safe and just space for humanity: Can we live within the doughnut? Oxfam Discussion Paper.
- Richter, B. D., Davis, M. M., Apse, C., & Konrad, C. (2012). A presumptive standard for environmental flow protection. *River Research and Applications*, 28(8), 1312–1321. https://doi.org/10.1002/rra.1511.
- Rockström, J., Edenhofer, O., Gaertner, J., & Declerck, F. (2020). Planet-proofing the global food system. *Nature Food*, 1, 3–5. https://doi. org/10.1038/s43016-019-0010-4.
- Röös, E., Bajželj, B., Smith, P., Patel, M., Little, D., & Garnett, T. (2017). Greedy or needy? Land use and climate impacts of food in 2050 under different livestock futures. *Global Environmental Change*, 47, 1–12. https://doi.org/ 10.1016/j.gloenvcha.2017.09.001.
- Rosa, L., Chiarelli, D. D., Tu, C., Rulli, M. C., & D'Odorico, P. (2019). Global unsustainable virtual water flows in agricultural trade. *Environmental Research Letters*, 14, 1–11. https://doi.org/10.1088/1748-9326/ab4bfc.
- Rosa, L., Rulli, M. C., Davis, K. F., Chiarelli, D. D., Passera, C., & D'Odorico, P. (2018). Closing the yield gap while ensuring water sustainability. *Environmental Research Letters*, 13, 1–11. https://doi.org/10.1088/1748-9326/aadeef.
- Rost, S., Gerten, D., Bondeau, A., Lucht, W., Rohwer, J., & Schaphoff, S. (2008). Agricultural green and blue water consumption and its influence on the

global water system. Water Resources Research, 44(9), 1-17. https://doi.org/10.1029/2007WR006331.

- Schader, C., Muller, A., El-Hage Scialabba, N., Hecht, J., Isensee, A., Erb, K. H., Smith, P., Makkar, H. P. S., Klocke, P., Leiber, F., Schwegler, P., Stolze, M., & Niggli, U. (2015). Impacts of feeding less food-competing feedstuffs to livestock on global food system sustainability. *Journal of the Royal Society Interface*, 12(113), 1–12. https://doi.org/10.1098/rsif.2015.0891.
- Schaphoff, S., Forkel, M., Müller, C., Knauer, J., von Bloh, W., Gerten, D., Jägermeyr, J., Lucht, W., Rammig, A., Thonicke, K., & Waha, K. (2018a). LPJmL4 – A dynamic global vegetation model with managed land – Part 2: Model evaluation. *Geoscientific Model Development*, 11(4), 1377–1403. https://doi.org/10.5194/gmd-11-1377-2018.
- Schaphoff, S., Von Bloh, W., Rammig, A., Thonicke, K., Biemans, H., Forkel, M., Gerten, D., Heinke, J., Jägermeyr, J., Knauer, J., Langerwisch, F., Lucht, W., Müller, C., Rolinski, S., & Waha, K. (2018b). LPJmL4 – A dynamic global vegetation model with managed land – Part 1: Model description. *Geoscientific Model Development*, 11(4), 1343–1375. https://doi.org/10. 5194/gmd-11-1343-2018.
- Schewe, J., Heinke, J., Gerten, D., Haddeland, I., Arnell, N. W., Clark, D. B., Dankers, R., Eisner, S., Fekete, B. M., Colón-González, F. J., Gosling, S. N., Kim, H., Liu, X., Masaki, Y., Portmann, F. T., Satoh, Y., Stacke, T., Tang, Q., Wada, Y., ... Kabat, P. (2014). Multimodel assessment of water scarcity under climate change. *Proceedings of the National Academy of Sciences*, 111(9), 3245–3250. https://doi.org/10.1073/pnas.1222460110.
- Schneider, U., Becker, A., Finger, P., Meyer-Christoffer, A., Ziese, M., & Rudolf, B. (2014). GPCC's new land surface precipitation climatology based on quality-controlled *in situ* data and its role in quantifying the global water cycle. *Theoretical and Applied Climatology*, 115, 15–40. https://doi. org/10.1007/s00704-013-0860-x.
- Shepon, A., Eshel, G., Noor, E., & Milo, R. (2018). The opportunity cost of animal based diets exceeds all food losses. *Proceedings of the National Academy* of Sciences of the United States of America, 115(15), 3804–3809. https://doi. org/10.1073/pnas.1713820115.
- Siebert, S., Kummu, M., Porkka, M., Döll, P., Ramankutty, N., & Scanlon, B. R. (2015). A global data set of the extent of irrigated land from 1900 to 2005. *Hydrology and Earth System Sciences*, 19(3), 1521–1545. https://doi.org/10. 5194/hess-19-1521-2015.
- Springmann, M., Clark, M., Mason-D'Croz, D., Wiebe, K., Bodirsky, B. L., Lassaletta, L., de Vries, W., Vermeulen, S. J., Herrero, M., Carlson, K. M., Jonell, M., Troell, M., DeClerck, F., Gordon, L. J., Zurayk, R., Scarborough, P., Rayner, M., Loken, B., Fanzo, J., ... Willett, W. (2018). Options for keeping the food system within environmental limits. *Nature*, 562, 519–525. https://doi.org/10.1038/s41586-018-0594-0.
- Steffen, W., Richardson, K., Rockström, J., Cornell, S. E., Fetzer, I., Bennett, E. M., Biggs, R., Carpenter, S. R., de Vries, W., de Wit, C. A., Folke, C., Gerten, D., Heinke, J., Mace, G. M., Persson, L. M., Ramanathan, V., Reyers, B., & Sörlin, S. (2015). Planetary boundaries: Guiding human development on a changing planet. *Science (New York, N.Y.)*, 347(6223), 1–10. https://doi.org/ 10.1126/science.1259855.
- Stenzel, F., Gerten, D., Werner, C., & Jagermeyr, J. (2019). Freshwater requirements of large-scale bioenergy plantations for limiting global warming to 1.5°C. *Environmental Research Letters*, 14, 1–13. https://doi.org/10.1088/ 1748-9326/ab2b4b.
- Tilman, D., Balzer, C., Hill, J., & Befort, B. L. (2011). Global food demand and the sustainable intensification of agriculture. *Proceedings of the National Academy of Sciences of the United States of America*, 108(50), 20260– 20264. https://doi.org/10.1073/pnas.1116437108.
- Tilman, D., & Clark, M. (2014). Global diets link environmental sustainability and human health. *Nature*, *515*, 518–522. https://doi.org/10.1038/nature13959.
- UN. (2015). Transforming Our World: The 2030 Agenda for Sustainable Development. A/RES/70/1. United Nations, New York, USA.
- Van Zanten, H. H. E., Meerburg, B. G., Bikker, P., Herrero, M., & de Boer, I. J. M. (2016). Opinion paper: The role of livestock in a sustainable diet: A land-use perspective. *Animal: An International Journal of Animal Bioscience*, 10(4), 547–549. https://doi.org/10.1017/S1751731115002694.
- Van Zanten, H. H. E., Herrero, M., Van Hal, O., Röös, E., Muller, A., Garnett, T., Gerber, P. J., Schader, C., & De Boer, I. J. M. (2018). Defining a land

boundary for sustainable livestock consumption. *Global Change Biology*, 24(9), 4185–4194. https://doi.org/10.1111/gcb.14321.

- Wada, Y., & Bierkens, M. F. P. (2014). Sustainability of global water use: Past reconstruction and future projections. *Environmental Research Letters*, 9, 1–17. https://doi.org/10.1088/1748-9326/9/10/104003.
- Wada, Y., van Beek, L. P. H., & Bierkens, M. F. P. (2012). Nonsustainable groundwater sustaining irrigation: A global assessment. *Water Resources Research*, 48(6), 1–18. https://doi.org/10.1029/2011WR010562.
- Wang, J., Liu, Q., Hou, Y., Qin, W., Lesschen, J. P., Zhang, F., & Oenema, O. (2018). International trade of animal feed: Its relationships with livestock density and N and P balances at country level. *Nutrient Cycling in Agroecosystems*, 110, 197–211. https://doi.org/10.1007/s10705-017-9885-3.
- Weindl, I., Bodirsky, B. L., Rolinski, S., Biewald, A., Lotze-Campen, H., Müller, C., Dietrich, J. P., Humpenöder, F., Stevanović, M., Schaphoff, S., & Popp, A. (2017). Livestock production and the water challenge of future food supply: Implications of agricultural management and dietary choices. *Global*

Environmental Change, 47, 121–132. https://doi.org/10.1016/j.gloenvcha.2017. 09.010.

- West, P. C., Gerber, J. S., Engstrom, P. M., Mueller, N. D., Brauman, K. A., Carlson, K. M., Cassidy, E. S., Johnston, M., MacDonald, G. K., Ray, D. K., & Siebert, S. (2014). Leverage points for improving global food security and the environment. *Science (New York, N.Y.)*, 345(6194), 325–328. https:// doi.org/10.1126/science.1246067.
- Willett, W., Rockström, J., Loken, B., Springmann, M., Lang, T., Vermeulen, S., Garnett, T., Tilman, D., DeClerck, F., Wood, A., Jonell, M., Clark, M., Gordon, L. J., Fanzo, J., Hawkes, C., Zurayk, R., Rivera, J. A., De Vries, W., Majele Sibanda, L., ... Murray, C. J. L. (2019). Food in the Anthropocene: The EAT-Lancet Commission on healthy diets from sustainable food systems. *Lancet (London, England)*, 393(10170), 447–492. https://doi.org/10.1016/S0140-6736(18)31788-4.
- Yin, X.-A., Yang, Z.-F., & Petts, G. E. (2011). Reservoir operating rules to sustain environmental flows in regulated rivers. *Water Resources Research*, 47(8), 1–13. https://doi.org/10.1029/2010WR009991.